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Denis Cotte on behalf of PS operations and supervisor team



PS Fault Breakdown 
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PLC for C200MHz communication lost

No beam from PSBooster

KFA21 and SMH57 faults

F16.QFO165 power supply

Magnet fault on ZT10.QDE01



Status of beams
Fixed target beams Status Comment

EAST Irrad/North
with nTOF Parasitic

Operational Several Accesses in IRRAD Area.
We try to give spills to the North branch

MTE Operational Delivered to SPS at ~7 ∙ 1012 ppp

TOF Operational Up to ~8 ∙ 1012 ppp (now at nominal Ip) 

AD Ready in PS ~1.4 ∙ 1013 ppp

LHC-type beams Status Comment

LHCPROBE, LHCINDIV Operational

LHC25 (12b, 72b) Operational

LHC25 BCMS (12b, 48b) Operational
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• Accumulated TOF intensity: ~11018 pot in 2018

nTOF proton on target (12/04/2018) 
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